Author’s response to reviews

Title: Agreement between patient-reported EQ-5D responses and EQ-5D responses mapped from the SF-12 in an injury population

Authors:

Emma McDermott (emma.mcdermott@monash.edu)
Pamela M Simpson (pamela.simpson@monash.edu)
Sarah Derrett (sarah.derrett@otago.ac.nz)
Shanthi Ameratunga (s.ameratunga@auckland.ac.nz)
Suzanne Polinder (s.polinder@erasmusmc.nl)
Ronan A Lyons (r.a.lyons@swansea.ac.uk)
Frederick P Rivara (fpr@u.washington.edu)
James E Harrison (james.harrison@flinders.edu.au)
Belinda J Gabbe (belinda.gabbe@monash.edu)

Version: 2
Date: 24 July 2014

Author’s response to reviews:

Apologies for the upload letter. This has been rectified now.